Author's response to reviews

Title: A global transcriptional view of apoptosis in human T-cell activation

Authors:
Min Wang (m-wang2@northwestern.edu)
Dirk Windgassen (dwindgassen@dendreon.com)
Eleftherios T Papoutsakis (e-paps@northwestern.edu)

Version: 5 Date: 21 October 2008

Author's response to reviews: see over
October 21, 2008

The BioMed Central Editorial Team

RE: Re-Submission of previously submitted manuscript MS: 8443998862070919 entitled: “A global transcriptional view of apoptosis in human T-cell activation” to BMC Medical Genomics.

To Whom It May Concern:

Thank you very much for the email on October 21, 2008 with the update of the status of our manuscript. We have carefully checked over our manuscript and have made changes for Figure 1 and Figure 4 (The time course is more accurately presented in the revised version). The revised manuscript is now submitted for your review.

Thank you very much for your attention.

Sincerely,

Eleftherios Terry Papoutsakis
Eugene DuPont Chair Professor
Dept. of Chemical Engineering & Delaware Biotechnology Institute
University of Delaware
15 Innovation Way
Newark, DE  19711
PH: 302-831-8376
FAX: 302-831-4841
EMAIL: epaps@udel.edu